Author's response to reviews

Title: Pregnant women at risk of primary infection with rubella virus in Burkina Faso

Authors:

Marc C Tahita (marctahita@yahoo.fr)
Judith M Hübschen (jJudith.huebschen@crp-sante.lu)
Zekiba Tarnagda (zekiba@hotmail.com)
Da Ernest (ernestda@yahoo.fr)
Emilie Charpentier (emilie.charpentier@crp-sante.lu)
Jacques R Kremer (jacques.kremer@education.lu)
Claude P Muller (claude.muller@crp-sante.lu)
Jean B Ouedraogo (jbouedraogo.irssbob@fasonet.bf)

Version: 2 Date: 17 December 2012

Author's response to reviews: see over
Dear Dr. Harris,

Please find enclosed our manuscript entitled

**Pregnant women at risk of primary infection with rubella virus in Burkina Faso**

for publication as a research article in BMC Infectious Diseases.

The manuscript provides important information about rubella circulation in Burkina Faso and the risk of infection during pregnancy. The potential impact of inclusion of rubella vaccine in the National Immunization Program is discussed.

The present manuscript has not been submitted or accepted for publication elsewhere. All authors have significantly contributed to the study and agree to the submission to BMC Infectious Diseases. The authors declare that there are no conflicting interests in relation to this study.

We are confident that you will find the manuscript suitable for publication.

Yours sincerely,

Prof. Dr. Claude P. Muller